Reviewer #1:
Scientific Quality: Grade A (Excellent)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (High priority)
Specific Comments to Authors: Very interesting case. A rare situation that may become more frequent

Thank you for your review.

Reviewer #2:
Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: An excellent reminder of a rare condition

Thank you for your review.

Reviewer #3:
Scientific Quality: Grade D (Fair)
Language Quality: Grade B (Minor language polishing)
Conclusion: Rejection
Specific Comments to Authors: This is a case report on a quite rare adverse event after PEG based bowel prep assumption, however it has already been described and I do not see nothing new in this report

Thank you for your opinion. I also read some case reports about PEG-induced anaphylaxis, but as a surgeon, I submitted this case report because I wanted to alarm the life-threatening complications of bowel preparation with PEG before colorectal cancer surgery.